Author's response to reviews

Title: Idiopathic toe walking and sensory processing dysfunction

Authors:

Cylie M Williams (cylie.williams@southernhealth.org.au)
Paul Tinley (ptinley@csu.edu.au)
Michael Curtin (mcurtin@csu.edu.au)

Version: 6 Date: 15 August 2010

Author's response to reviews:

Formatting changes to be made:
-------------------------------
Major revisions (we require the author to make these changes)

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Currently your references are cited out of order - Reference order has been checked and modified. If not accurate please clarify if table references are to be included.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you.

Cylie